

Temi di discussione

(Working Papers)

The impact of the IRB approach on the relationship between the cost of credit for public companies and financial market conditions

by Raffaele Gallo

I1290



Temi di discussione

(Working Papers)

The impact of the IRB approach on the relationship between the cost of credit for public companies and financial market conditions

by Raffaele Gallo

Number 1290 - July 2020

The papers published in the Temi di discussione *series describe preliminary results and are made available to the public to encourage discussion and elicit comments.*

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Monica Andini, Audinga Baltrunaite, Marco Bottone, Davide Delle Monache, Sara Formai, Francesco Franceschi, Salvatore Lo Bello, Juho Taneli Makinen, Luca Metelli, Mario Pietrunti, Marco Savegnago. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

THE IMPACT OF THE IRB APPROACH ON THE RELATIONSHIP BETWEEN THE COST OF CREDIT FOR PUBLIC COMPANIES AND FINANCIAL MARKET CONDITIONS

by Raffaele Gallo^{*}

Abstract

This paper examines whether the regulatory approach adopted by banks to calculate capital requirements has a different impact on the loan rates for public and private companies when financial market conditions change. Using Italian data for the period 2008-18, the analysis documents that the adoption of the internal ratings-based (IRB) approach has led to a significantly greater sensitivity of the loan rates applied to public companies to financial market conditions, proxied by the VSTOXX index. For credit granted by IRB banks, being public is associated with a significant loan cost advantage when the level of financial instability is low. However, when VSTOXX rises, public companies experience a greater increase in loan rates than private firms; the effect is determined mostly by less capitalized IRB banks. In contrast, for credit granted by banks that adopt the standardized approach (SA), public borrowers do not benefit from a significant loan cost advantage compared with private ones, and a change in financial market conditions has a similar impact on loan rates for both types of companies.

JEL Classification: G01, G20, G21, G32.

Keywords: credit risk regulation, public firm, financial stability, interest rates, bank credit. **DOI**: 10.32057/0.TD.2020.1290

1. Introduction	
2. Review of the regulatory background and related research	7
3. Research hypotheses	
4. Data and sample	
5. Methodology and descriptive statistics	
6. Results	
7. Robustness checks	
8. Conclusions	
Appendix	
References	
Tables and figures	
5	

Contents

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research.

1. Introduction¹

After the introduction of the Basel II framework in 2007, banks are allowed to choose between two main regulatory approaches to calculate capital charges for credit risk. Under the standardized approach (SA), risk weights associated with each exposure are fixed and constant over time. Under the internal ratings-based approach (IRB), risk weights depend on banks' internal risk models validated by the supervisors. When using the latter method, therefore, banks calculate capital charges relying on their own estimate of the risk associated with each exposure in their portfolio.

The use of internal models has represented a major change in the regulatory assessment of credit risk. The implications associated with the adoption of the IRB approach are at the core of supervisory scrutiny and of the debate on the next regulatory framework (BCBS, 2017).

Since capital charges are linked to asset risk under the IRB approach while they are fixed over time under SA, the literature has underlined that the regulatory capital is more risk-sensitive for IRB than for SA banks (Behn et al., 2016b; Kashyap and Stein, 2004; Repullo and Suarez, 2004, 2013). Consequently, capital requirements for IRB financial institutions may significantly change over time, rising after a negative shock to borrower's riskiness.

In addition, the IRB approach allows exerting greater discretion in the assessment of the borrower's creditworthiness because IRB banks can estimate the risk of each exposure with their own models. However, the literature has documented that the discretion is lower when there is an external benchmark that can be compared with banks' internal evaluations (Behn et al., 2016a; Firestone and Rezende, 2016; Plosser and Santos, 2014).

The abovementioned characteristics of the IRB approach, i.e. sensitivity to risk and reliance on external benchmarks, may have an asymmetric impact on the cost of loans granted to public and private borrowers. For a public borrower, the market price of the firm's share is a benchmark for IRB banks' evaluations; in contrast, for a private firm, IRB lenders' internal estimates cannot be compared with an external benchmark specifically related to the same borrower. Since changes in the market-based benchmark value affect IRB banks' evaluations for public firms, the cost of IRB loans may be more sensitive to financial market conditions for public borrowers than for private ones. Therefore, after a rise in financial instability, public firms may experience an increase in IRB loan rates greater than that faced by private companies. Instead, for SA loans, changes in financial market conditions may not affect loan rates for public and private borrowers differently.

¹ I thank F. Ciocchetta, S. De Mitri, R. Felici, and M. Moscatelli for their support in data collection. The work benefited from the useful comments of N. Branzoli, L. Carpinelli, G. Cascarino, F. Columba, D. Drago, G. Gobbi, G. Guazzarotti, A. Ilari, D. Loschiavo, S. Magri, and F. Palazzo. I also thank several colleagues in the Bank of Italy's Directorate General for Economics, Statistics and Research and seminar participants at the Bank of Italy's Workshop "Banking Research Network" (Rome, September 2019) for their useful suggestions.

This effect is relevant for the literature on the cost and benefits of going public. Several studies have documented that public firms benefit from a significant loan cost advantage with respect to private companies because they can rely on lower costs of information production and greater bargaining power (Pagano et al., 1998; Saunders and Steffen, 2011). The adoption of the IRB approach may reduce the loan cost advantage of listing in periods of financial market instability.

The aim of this paper is to evaluate whether the regulatory approach adopted by Italian banks to calculate capital requirements for credit risk affects the sensitivity of public and private firms' borrowing costs to changes in financial market conditions.

The sample analyzed is made of public and private borrowers with at least one loan granted respectively by a SA bank and an IRB bank in the same quarter, between 2008 and 2018. This identification strategy allows examining the loan rate applied to the same firm by at least two banks that adopt a different regulatory approach.

The empirical analysis indicates that public firms benefit on average from a loan cost advantage of 73 basis points (bps) with respect to private companies for credit granted by IRB banks. However, the adoption of the IRB approach leads to a greater sensitivity of the loan rates applied to public firms to financial market conditions, measured by the VSTOXX index. After a rise in VSTOXX of one standard deviation (about 8 units), public firms experience an increase in loan rates 31 bps greater than that faced by private companies. In contrast, for credit granted by SA banks, being listed is not associated with a significant loan cost advantage and a change in financial market conditions affects interest rates for public and private borrowers in a similar way.

The evidence concerning IRB banks may depend on the level of their capital ratios: greater capital buffers may reduce the loan rate sensitivity to financial market conditions. Even if IRB banks with different capital ratios assign the same riskiness to a public borrower, less capitalized IRB financial institutions may have a stronger reaction to a credit risk shock because they find it more problematic to comply with an increase in regulatory capital requirements (Behn et al., 2016b). Consequently, they may raise loan rates for public firms more than other IRB banks in reaction to an increase in financial instability. The results confirm that the reduction in the loan cost advantage of public firms associated with a deterioration in financial market conditions is more pronounced if the IRB bank has less capital.

The main findings hold also after considering lender bank and firm heterogeneity and using a matched sample that allows comparing borrowers with statistically similar characteristics.

This work contributes to two main strands of the extant literature. First, this paper expands the analysis of the impact of the IRB approach on bank lending (Behn et al., 2016a, 2016b; Berg and Koziol, 2017; Plosser and Santos, 2014; Repullo and Suarez, 2004, 2013). Second, it is related to

the literature on the loan cost advantage of public firms (Pagano et al., 1998; Saunders and Steffen, 2011) and on the effect of exogenous shocks on this benefit (Gallo, 2019; Santos and Winton, 2008). To the best of my knowledge, this paper is the first to assess the impact of the IRB approach on the relationship between the loan cost advantage of being public and the financial market climate.

The remaining part of this paper is organized as follows. Section 2 reviews the related literature and outlines the regulatory background. Section 3 presents the research hypotheses. Section 4 describes the dataset. Section 5 discusses the methodology used in the analysis. The main results are reported in Section 6. Section 7 contains robustness checks, while Section 8 concludes.

2. Review of the regulatory background and related research

2.1. The introduction of the model-based regulation

The entry into force of the Basel II framework in 2007 has significantly modified the credit risk assessment of financial institutions. The regime of Basel I classifies each bank asset in a list of predetermined buckets and assigns a fixed risk weight to each category. In contrast, the Basel II framework allows establishing a stronger link between capital charges and the actual risk of assets (Behn et al., 2016b). This framework allows banks to choose between the standardized approach (SA) and the internal ratings-based approach (IRB) for calculating capital requirements. In both methods capital requirements are determined in terms of risk-weighted assets (RWA), which are estimated by multiplying each risk weight with the value of the corresponding asset. The Basel III framework has confirmed this distinction.

The standardized approach is substantially similar to that prescribed under the Basel I framework (i.e. fixed risk weights assigned to each bucket of assets). For assets rated by a specialized agency, the risk weight can change after a downgrade or an upgrade of the external credit rating; while, from a regulatory point of view, the riskiness of unrated assets remains constant over time. Under SA, all uncollateralized loans granted to unrated firms have a risk weight of 100 per cent.

Under the IRB approach, the risk weight of each asset depends on banks' internal risk models. The determinants of asset risk estimates are four parameters: the probability of default (PD), the loss given default (LGD), the exposure at default (EAD), and the effective maturity of the loan.² Each model has to be approved and validated at least once a year by regulators. The estimates under

² Banks estimate all four parameters under the advanced IRB (AIRB) approach, while they estimate only the PD in the foundation IRB (FIRB) approach and fixed standard values are assumed for other parameters. The Italian IRB banks have almost exclusively adopted the AIRB method; therefore, this distinction does not affect the main findings.

IRB have to be periodically updated. Consequently, risk weights and related capital charges for each exposure may change over time.

Banks are incentivized to adopt the IRB approach because capital requirements are substantially lower under IRB than under SA, but the adoption is associated with substantial administrative costs and organizational efforts (Behn et al., 2016b). Therefore, only few large banks have adopted the IRB approach.

2.2. The main implications of the IRB approach

The financial literature has explored two main implications of the introduction of the IRB approach. First, several studies have underlined that the IRB risk-based requirements have raised the sensitivity of bank regulatory capital to asset risk compared with the flat requirements of SA (and Basel I) (Repullo and Suarez, 2004). A potential issue associated with the introduction of this regulation is that it may tend to amplify business (or financial) cycle fluctuations (Behn et al., 2016b; Kashyap and Stein, 2004; Repullo and Suarez, 2013) and to exacerbate financial instability (EBA, 2013). If asset risk measures are responsive to a negative shock, then capital requirements will rise after an increase in the borrower's riskiness. Banks that are unwilling or unable (e.g. less capitalized) to raise new equity may be forced to deleverage, exacerbating the negative impact of the shock. However, the incentive to reduce the loan portfolio riskiness when credit conditions deteriorate may enhance bank safety, and, consequently, the solvency of the banking system in the long-term (Repullo and Suarez, 2013).

Second, the literature finds evidence of a significant variability across IRB banks of RWA densities³ (Le Leslé and Avramova, 2012; Mariathasan and Merrouche, 2014; Pérez Montes et al., 2018) and of risk-weights assigned to the same borrower (Behn et al., 2016a; Berg and Koziol, 2017; Plosser and Santos, 2014). Some degree of idiosyncratic variation in internal estimates is allowed and encouraged: if banks' risk models are too similar, then all banks would be doing the same action at the same time, raising endogenous risk concerns (Berg and Koziol, 2017). Moreover, the high complexity of internal risk models and the adoption of different business models make risk estimates comparison difficult across financial institutions (Cannata et al., 2012; Cucinelli et al., 2018). However, controlling for several confounding factors, the literature suggests that the risk estimates (RWA density and risk-weights) of IRB banks with lower capital on average are more downward biased than those of more capitalized IRB banks (Behn et al., 2016a; Plosser and Santos, 2014). This result is consistent with a strategic risk-modelling hypothesis, which consists of a systematic underestimation of asset risk by IRB banks to reduce capital requirements.

³ The RWA density is defined as the ratio of risk-weighted assets to total assets.

As underlined by Behn et al. (2016a), the variability in IRB banks' risk estimates is lower when there is an external benchmark that can be compared with banks' internal estimates because misreporting would be more likely to be detected by the supervisors. Firestone and Rezende (2016) document that banks disagree less on their estimation of LGDs when they evaluate more transparent borrowers, such as public and rated firms. Plosser and Santos (2014) find that IRB banks, mainly those with lower capital, report on average downward biased estimates and that this bias is concentrated on the riskiness of private firms. For these borrowers, in fact, banks have greater discretion as to the inputs of their risk models.

2.3. The effects of being public

The decision of going public has relevant implications for firms. Public companies obtain better and cheaper access to external equity capital (Brav, 2009). Listed firms have lower costs of information production and greater bargaining power, they can therefore rely on a significant cost advantage in the loan market with respect to private companies (Pagano et al., 1998; Saunders and Steffen, 2011).

A strand of literature has examined the effect of exogenous shocks on the loan cost difference between public and private borrowers. Santos and Winton (2008) find that the increase in syndicated loan spreads during recessions is significantly lower for listed firms with public debt market access than for other listed companies. This evidence is consistent with the hypothesis that banks hold an informational monopoly that allows them to increase interest rates mostly to more opaque borrowers in contexts characterized by high information asymmetry (Rajan, 1992), such as in recessions. Gallo (2019) shows that the cost advantage of public firms in the syndicated loan market declines during high volatility periods because a rise in financial instability weakens their bargaining power and the information benefits of being listed.

Both empirical studies examine a sample of syndicated loans. However, the results obtained by analyzing these contracts might not be extended to the bilateral credit market. Since multiple syndicate participants charge a unique rate to the borrower, it is difficult to assess to which extent the characteristics of each bank (i.e. the regulatory approach adopted) affect loan pricing. Moreover, the presence of multiple lenders weakens the strength of the bilateral relationship between the borrower and a specific bank. Finally, a syndicated loan may not absorb regulatory capital of the lenders that decide to sell their shares of the loan in the secondary market. As a result, the impact of the regulatory approach adopted by each syndicate participant on loan rates may be more difficult to evaluate. In contrast, the adoption of a sample of bilateral loan contracts allows assessing how the

regulatory approach of the bank affects the cost of credit, controlling for the characteristics of the lender and the strength of bank-firm relationships.

Finally, the literature has also focused on the decision of several firms to remain private or to delist from a stock exchange (Bharath and Dittmar, 2010; Doidge et al., 2017). Among potential reasons, some companies may be interested in being more opaque and in decreasing outside scrutiny (Leuz et al., 2008; Marosi and Massoud, 2007). Since outside agents can rely on market prices continuously updated, each action of public firms is scrutinized and the reassessment of their value is easier than that of private companies. These studies suggest that the presence of an external benchmark may significantly affect firms' decisions.

3. Research hypotheses

The contributions mentioned in Section 2.2 indicate that IRB banks' evaluations on the creditworthiness of public firms may be closely related to borrowers' share prices.⁴ Since internal estimates are risk-sensitive for IRB banks, a financial shock that negatively affects the stock market may be directly incorporated in IRB banks' evaluations of public firms, indicating an increase in their riskiness. As a result, IRB banks might raise the interest rates applied to public firms after a worsening of financial market conditions.

In contrast, lenders cannot rely on a firm-specific market benchmark for evaluating private borrowers. Therefore, the interest rates applied by IRB banks to private firms may be not or only weakly correlated to financial market conditions.

Finally, under SA, risk weights associated with credit exposures both to public and private borrowers do not strictly depend on financial market conditions because capital charges are determined when the loan is made and do not change (Behn et al., 2016b).⁵ Consequently, the estimates of SA banks on the public firms' creditworthiness may be significantly less sensitive to a change in financial instability than those of IRB banks.

Overall, the adoption of the IRB approach may have a significant impact on loan pricing. For credit granted by IRB banks, the loan cost advantage of being public may decrease when financial market conditions deteriorate. In contrast, for SA loans, a change in the financial market climate may not affect the cost of credit for public and private borrowers differently. Therefore, the first testable hypothesis is:

⁴ For example, several banks adopt a Merton-based methodology to evaluate public corporates (e.g. Barclays, 2015).

⁵ Under SA, the risk weight can change over time for firms rated by credit rating agencies. Unreported analyses show that the main findings hold also controlling for the presence of rated firms in the examined sample.

H1. A deterioration in financial market conditions leads to an increase in IRB loan rates greater for public firms than for private companies, while for SA banks it has a similar impact on the cost of loans applied to public and private borrowers.

Even if the internal models of IRB banks with different capital ratios indicate the same increase in capital charges, financial institutions may react by applying different loan rates. Less capitalized banks may have a stronger reaction to an increase in capital requirements because raising capital may be more problematic for them (Behn et al., 2016b). When capital requirements related to exposures to public firms rise after a financial shock, IRB banks with lower capital ratios may raise interest rates more than other IRB banks. Consequently, the reduction in the loan cost advantage of public firms associated with a worsening of the financial market climate (H1) may be lower for credit granted by more capitalized IRB banks. The second hypothesis tested in this paper is therefore that:

H2. The loan rates applied to public firms by less capitalized IRB banks are more sensitive to financial market conditions than those applied by more capitalized IRB banks.

4. Data and sample

4.1. Data sources

The sample consists of quarterly data on credit to Italian non-financial firms from the Italian Credit Register ("Centrale dei Rischi", CR) from the first quarter of 2008 to the second quarter of 2018. CR is maintained by the Bank of Italy and covers the population of individual borrowers' outstanding exposure of over \notin 30,000 with a single intermediary.⁶ For each exposure, the database provides detailed information on the lender, the borrower identity, and the type of credit (credit lines, term loans, and loans backed by accounts receivable). Data on the interest rates applied by banks are obtained from a section of the CR, TAXIA, which contains information on the interest rate charged on all loans granted by a representative sample of Italian banks.⁷

This work focuses on credit lines. Since rates on this source of financing are highly standardized among banks, they are more comparable than those on other credit types (Sette and Gobbi, 2015). In contrast to term loans, for example, credit lines do not have a specific maturity, are granted for no specific purpose, and are typically not backed by collateral.⁸ The lender can modify quickly and unilaterally the contract terms, including prices. In addition, credit lines represent a significant source of financing for firms, mainly during crisis periods (Acharya et al., 2014).

⁶ For CR, the reporting threshold was €75,000 before 2009.

⁷ The credit granted by this sample of banks accounts for more than 80 per cent of total bank lending in Italy. For TAXIA, the reporting threshold is \notin 75,000.

⁸ The results are robust to using also interest rates on term loans and on loans backed by account receivables.

The baseline model considers interest rates on revolving credit lines inclusive of fees and commissions. Therefore, this measure takes into account the overall cost of credit charged to borrowers for each loan.⁹ The interest rate for each credit line is estimated by dividing the amount due, including fees and commissions, by the amount of loans, multiplied by the days this amount was outstanding.

Firm accounting data are retrieved from the database of Cerved group, which is a leading information provider in Italy, while data on banks' balance sheets are obtained from Supervisory Reports of Bank of Italy. Information on securities issued by each firm is collected from the Securities Database of Bank of Italy; other financial variables are obtained from Datastream. Finally, the regulatory approach adopted by each Italian bank is singled out by relying on confidential supervisory information and publicly available banks' Pillar 3 reports.

In addition, the analysis also considers information on bank-firm relationships, which have significant effects in the Italian credit market (Sette and Gobbi, 2015).¹⁰ Long relationships allow banks to accumulate more information about borrowers' creditworthiness over time (Boot, 2000). Similarly, holding a large share of the overall credit granted to the firm allows lenders to have better access to significant information (Elsas, 2005). As a result, borrowers may benefit from the reduction in information asymmetry associated with close relationships (long duration or a large share of credit). However, since a stronger relationship may indicate a greater hold-up power of lenders, it may imply also greater costs for borrowers in switching to different lenders (Ioannidou and Ongena, 2010; Rajan, 1992).¹¹ To take into account the potential effects of bank-firm relationships, information on the duration of the relationship and on the share of total credit for each bank-firm combination are also included in the estimations.

4.2. Sample construction

The sample is built as follows. First, credit is aggregated at the banking group level because lending and funding policies are typically decided at this level.

Second, the following loans are excluded: loans to firms with no balance sheet information in the Cerved database; non-performing loans; loans with a missing or nearly zero gross interest rates (less than 0.01 per cent); undrawn credit lines (drawn-to-granted amount ratio lower than 1 per cent); and

⁹ The results obtained by adopting net interest rates are qualitatively similar to those presented in the following estimates.

¹⁰ The analysis controls for mergers and acquisition among banks. If a firm had a relationship with a bank and the bank is acquired or merged, the relationship is considered as still existing with the newly constituted financial institution.

¹¹ The empirical evidence on the impact of stronger relationships on interest rates is mixed. For example, Berger and Udell (1995) and Bharath et al. (2011) find that the strength of relationships and interest rates are negatively related, while Degryse and Ongena (2005) and Ioannidou and Ongena (2010) show that interest rates rise when the relationship is stronger.

loans with extremely high interest rates.¹² To reduce the risk that outliers could affect results, the interest rate variable is also trimmed at the 5th and 95th percentile.

Third, the sample is restricted to firms with at least one loan granted respectively by a SA and an IRB bank in the same quarter. This allows comparing how IRB and SA lenders evaluate the same borrower, overcoming potential identification issues.

Fourth, a potential shortcoming of this analysis is that public firms are a small fraction of the overall number of borrowers.¹³ In addition, as documented in the literature (Section 2.3), public firms are significantly different from private companies, both in terms of financial indicators (e.g. lower leverage ratios) and of unobservable characteristics (e.g. quality of management and governance). Therefore, these differences may impair the comparison between the borrowing costs of public and private firms. Previous studies on the loan cost advantage of being public have mainly adopted a sample of syndicated loans (e.g. Saunders and Steffen, 2011), which are typically granted to a subset of larger (public and private) firms. Other related studies focused on the impact of the initial public offering (IPO) on bank loan costs by comparing a sample of private firms that go public with other companies that remain private (e.g. Pagano et al., 1998).

To ensure comparability between public and private firms, two alternative sample specifications are employed. First, in the baseline model, the sample is restricted to firms with access to market financing (i.e. stock and bond markets). Since only few firms with a particular set of observable and unobservable characteristics have access to stock and bond markets (Denis and Mihov, 2003; Hale and Santos, 2008), this restriction allows examining more similar firms.

Companies listed in the stock market are considered as "public borrowers", while unlisted firms with access to the bond market are considered as "private borrowers". I define the firms with at least one outstanding bond in t as companies with access to the bond market.¹⁴ These companies represent the control group of the analysis. Indeed, banks cannot rely on timely market information on firms with privately placed bonds because there are not publicly disclosed secondary market prices for these securities. Moreover, given that the liquidity in the secondary bond market of Italian firms is significantly low,¹⁵ also public bond prices could not be a reliable external benchmark for

¹² Fees and commissions are applied on credit granted, while net interest rates in CR are estimated on the basis of the usage of the credit line. Consequently, if a credit line is used for a relatively small amount or for a short period, fees and commissions are larger than net interest rates. This leads to extremely large gross interest rates.

¹³ Public companies are 0.2 per cent of the examined sample of borrowers, but they receive 5 per cent of the total credit granted in the analyzed period.

¹⁴ The results are robust to considering in this category firms that have issued bonds in one, three, or ten years before *t*.

¹⁵ Accornero et al. (2018) document that the Italian corporate bond market is significantly smaller than those of France,

UK, and US.

banks. Consequently, also companies with public bond market access are included in the control group of private borrowers.¹⁶

The second alternative sample restriction relies on a matching procedure between listed firms and the full sample of unlisted borrowers. This method, described in Section 7.4, allows comparing firms with statistically similar credit risk, proxied by accounting-based (observable) characteristics.

Overall, the sample employed in the following analyses includes 30728 quarterly credit relationships (7318 loans to public firms, and 23410 to private companies) between 766 non-financial firms (188 public and 578 private borrowers) and 137 Italian banking groups.

With regard to the regulatory approach of lenders, 11 Italian banks have adopted the IRB approach in the analyzed period. The credit granted by IRB banks accounts for about 42 per cent of total bank lending granted to the borrowers included in the sample. Table 1 reports descriptive statistics of the bank-firm relationships included in the sample, indicating the shares of relationships with IRB banks and of the amount of revolving credit lines granted by IRB banks for public and private firms. Compared with private companies, public firms establish on average fewer relationships with IRB banks (41 versus 45 per cent) but they receive a greater share of credit from them (57 versus 52 per cent). Overall, both IRB and SA banks play a significant role for both categories of borrowers. This allows excluding that public or private companies mainly rely on a specific type of bank.

4.3. The measure of financial market instability

The VSTOXX index is the main measure of the level of financial market instability used in the paper. This index reflects the market expectations of equity volatility across all EURO STOXX 50 options over the next thirty days. By considering the implied volatility, a forward-looking perspective is adopted, more consistent with that of lenders. In addition, the use of a European volatility index allows employing a measure that is less affected by the idiosyncratic stock volatility of each Italian firm and signals more accurately the climate of international financial markets. For Italian non-financial firms, the significant rises in financial instability observed during the examined period (2008-2018) were largely unexpected and exogenous events. These elements allow improving the identification of the impact of financial instability on firms' borrowing costs.

Figure 1 shows the quarterly time series of the VSTOXX index over the investigated period. The volatility index exhibits large upward swings above the mean value equal to 24 (the median value is

¹⁶ The reported findings remain unchanged by including firms with public bond market access in the group of public firms or by excluding these companies from the sample. An unreported robustness check documents that an increase in financial market instability does not affect loan rates for firms with public bonds differently than those applied to companies with privately placed bonds.

22), mainly during the great financial crisis and the sovereign debt crisis; the quarters after these crisis periods were characterized by lower volatility. Therefore, the presence of low and high volatility periods over the examined time interval constitutes an ideal setting for this analysis.

As a robustness check, qualitatively similar results are obtained by replacing VSTOXX with the historical volatility of the FTSE MIB index (the Italian main stock market index), estimated as the standard deviation of FTSE MIB index returns over the year preceding *t*.

5. Methodology and descriptive statistics

To test hypothesis *H1*, the following model is estimated for loans granted by SA and IRB lenders, separately:

$$Rate_{ijt} = \beta_1 Public_{it} + \beta_2 Public \cdot VSTOXX_{it} + \gamma Relationship_{ijt} + \eta Borrower_{it-1}$$
(1)
+ $\delta_{jt} + \varepsilon_{ijt}$

The dependent variable, *Rate*, is the interest rate on revolving credit lines granted by bank j to firm i on quarter t. *Public* is a dummy variable equal to 1 if the firm i is listed in t and to 0 otherwise. In line with previous studies mentioned in Section 2.3, I expect to find a negative coefficient of this variable both for SA and for IRB loans.

The main variable of interest, *Public*·*VSTOXX*, is the interaction between *Public* and *VSTOXX*, which is equal to the value of the VSTOXX index at the end of the quarter t-1.¹⁷ Consistent with *H1*, the coefficient of this variable for IRB loans should be significant and positive. In contrast, the coefficient of *Public*·*VSTOXX* is expected to be not significant for loans granted by SA banks.

The main model specification also includes bank-quarter fixed effects (δ_{jt}), which control for bank-level unobserved heterogeneity in each quarter. They also absorb the impact of the variable *VSTOXX*, which therefore does not appear in Eq. (1).

Moreover, two vectors of control variables are added. First, the potential effects of bank-firm relationships (vector *Relationship*) is taken into account by introducing two variables: *Share*, the share of total credit (revolving credit lines, term loans, and loans backed by accounts receivables) granted by bank j to firm i in t; *RelDuration*, a dummy equal to 1 if the bank-firm relationship started at least in the year before t. A higher value of both variables indicates stronger bank-firm relationships.

Second, the borrower's characteristics are controlled by adding the vector *Borrower* in Eq. (1). It includes firms' accounting data (*Size*, *Leverage*, *Fixed Assets*, *EBITDA*, *IntCovRatio*) as well as

¹⁷ The results are robust to adopting alternatively the value of the VSTOXX index at the end of the quarters t-2, t-3, and t-4.

Industry Dummies (indicator variables for the industry of firm *i* based on 2-digit ATECO).¹⁸ This vector takes also into account a measure of the borrower's credit risk, *HighRisk*, estimated as a dummy variable equal to 1 if the credit score assigned by Cerved to firm *i* in the year preceding *t* is greater than six and 0 otherwise.¹⁹

The model described in Eq. (2) is estimated to test hypothesis H2:

$$Rate_{ijt} = \beta_{1}Public_{it} + \beta_{2}Public \cdot VSTOXX_{it} + \beta_{3}Public \cdot HighBankCapital_{ijt}$$
(2)
+ $\beta_{4}Public \cdot VSTOXX \cdot HighBankCapital_{ijt} + \gamma Relationship_{ijt}$
+ $\eta Borrower_{it-1} + \delta_{jt} + \varepsilon_{ijt}$

This model also includes the variable HighBankCapital, a dummy variable equal to 1 if the bank j has a capital ratio above the 75th percentile²⁰ of the distribution of capital ratios²¹ in the year preceding t and 0 otherwise. Given the inclusion of bank-quarter fixed effects, the variable HighBankCapital does not appear in Eq. (2). The main focus is the triple interaction Public VSTOXX HighBankCapital. Consistent with H2, this variable is expected to have a significant and negative impact on interest rates only for loans granted by IRB banks.

Table A.1 in the Appendix presents the complete list of variables with their relative sources.

5.1. Descriptive statistics

Table 2 shows a set of descriptive statistics of public and private borrowers included in the sample (i.e. firms with access to market financing). Consistent with the literature on the loan cost advantage of public and private companies (Section 2.3), the average interest rate is lower for public borrowers than for private ones.

Public firms are significantly larger and have lower leverage than private companies. However, private borrowers included in the sample have better credit scores and greater accounting ratios. As regards the vector *Relationship*, private companies have slightly stronger bank-firm relationships, examining both the share of total credit and the duration. The variable *RelDuration* indicates that new bank-firm relationships are less than 10 per cent of the loans included in the sample, both for

¹⁸ Unreported analyses show that the results are robust to controlling for additional variables, such as the number of financing banks, the ratio of revolving loans to total loans at the bank-firm level, and a set of indicator variables for geographical area of the country.

¹⁹ The credit score is an indicator of the probability of default that is computed annually by Cerved, following the Z-score methodology (Altman et al., 1994). The Z-score varies from 1 (safest) to 9 (riskiest). Therefore, following Sette and Gobbi (2015), the variable *HighRisk* indicates the firms with a Z-score above the median.

²⁰ The choice of this threshold is due to the significant asymmetry in the distribution of Italian bank capital ratios. Few banks have significantly low or high capital ratios, while the others have ratios close to the median. Consequently, the identification of a significant heterogeneity among banks requires the adoption of a relatively high threshold. The results are robust to considering higher threshold values.

²¹ The capital ratio is estimated as the ratio of total regulatory capital to risk-weighted assets.

public and private borrowers. The characteristics of control variables remain constant both in low and high volatility periods.

The abovementioned differences between public and private borrowers motivate the inclusion of controlling variables in the main model and the adoption of a propensity score matching technique (Section 7.4).

6. Results

6.1. The impact of the IRB approach on loan rates for public and private firms

Table 3 reports the results obtained from the estimation of Eq. (1).²² In column (1), the estimates obtained by examining the sample of SA loans show that the coefficient of *Public* is not significant, suggesting that the interest rates applied to public companies by SA banks are not significantly different from those charged to private firms. Probably, since SA banks adopt less sophisticated methods to evaluate the creditworthiness of their borrowers and they may rely more heavily on accounting measures. As a result, for credit granted by SA banks, the loan cost advantage of being public is not significant once borrower characteristics have been included. As regards the variable of interest, the interaction *Public·VSTOXX* does not have a significant impact on interest rates, consistent with *H1*.²³

In contrast, the results presented in column (2) of Table 3, obtained with the sample of IRB loans, show that there is a clear-cut benefit for public firms: their cost of credit is on average 73 bps lower compared with private companies, as suggested by the coefficient of *Public*. Moreover, the interaction *Public-VSTOXX* has a positive and significant coefficient. As predicted by *H1*, this result implies that the loan cost advantage of being public depends on financial market conditions. An increase in VSTOXX of one standard deviation (about 8 units) raises interest rates by 31 bps more for public firms than for private companies.²⁴ Consequently, an increase of about 19 units in VSTOXX may potentially nullify the loan cost advantage of public borrowers.²⁵ At the same time, the results of this model imply that a reduction in financial instability raises the loan cost advantage of being public with the same magnitude.

As regards the effect of bank relationships, the coefficients of the two measures included in this vector (i.e. *Share* and *RelDuration*) have an opposite sign in both columns, implying that these variables might capture slightly different effects. The negative sign of *Share* indicates that a greater

²² The following tables report robust standard errors. The results are robust to clustering at the firm level, the bank level, and the bank-firm level.

 $^{^{23}}$ As anticipated in Section 5, since the model includes bank-quarter fixed effects, it focuses on differential effects; while the level effect of *VSTOXX* is not estimated.

²⁴ Using the estimates in column (2): $8 \cdot 0.039 = 0.312$.

 $^{^{25}}$ Using the estimates in column (2): 0.731/0.039 = 18.74.

share of total credit held by the bank leads to a reduction in interest rates. This result may suggest the existence of evergreening practices. Indeed, banks may grant relatively "cheap" credit to a borrower to which they have a significant exposure in order to postpone the accounting of credit losses (Albertazzi and Marchetti, 2010). In contrast, the positive sign of *RelDuration* implies that longer bank-firm relationships are associated with greater loan rates, suggesting that banks may apply lower interest rates to new customers for commercial purposes. This result is consistent with the findings of Ioannidou and Ongena (2010), who find that firms switching to new banks obtain better contract terms.²⁶

Other control variables have the expected signs, consistent with the literature. Both SA and IRB banks apply lower interest rates to borrowers with greater size, fixed assets, higher EBITDA, and better credit scores. Higher *Leverage* is associated with greater interest rates only in the model estimated by employing SA loans, while the interest coverage ratio (*IntCovRat*) does not significantly affect the cost of loans in both models when the credit score dummy (*HighRisk*) has been included.

Finally, column (3) of Table 3 presents the results obtained by employing the full sample (both IRB and SA loans) and by interacting *Public*·*VSTOXX* with *IRB* (a dummy variable equal to 1 if the bank *j* adopts the IRB approach in *t* and 0 otherwise).²⁷ The estimates obtained by including the triple interaction confirm previous findings.

Overall, the tests reported in this section are consistent with *H1* and confirm that the IRB approach has significantly raised the sensitivity of the loan cost advantage of public firms to financial market conditions.

Unreported analyses also verified whether the adoption of the IRB approach has an impact on the credit amount granted to public and private borrowers. This possibility is explored by using the quarterly change in the logarithm of the amount of total credit (revolving credit lines, term loans, loans backed by accounts receivables) granted by bank *j* to firm *i* as the dependent variable of Eq. (1). The coefficient of the interaction *Public*·*VSTOXX* is not statistically significant both for IRB and for SA loans.²⁸ These findings may suggest that IRB banks react to a worsening of financial market conditions mainly by adjusting interest rates instead of reducing credit quantity, at least in the short-term.

²⁶ An unreported robustness check shows that the main findings remain unchanged also by interacting the *Relationship* variables with *VSTOXX*. This additional analysis documents that the level of financial instability does not affect the impact of relationship banking variables on loan rates in both subsamples.

²⁷ The variable *IRB* does not appear in the model because bank-quarter fixed effects absorb its impact.

²⁸ If Eq. (1) is estimated by using the quarterly change in the logarithm of the amount of revolving credit lines (instead of total credit) as the dependent variable, the results indicate that the credit quantity granted by IRB banks decreases relatively more for public borrowers than for private firms. However, the effect is economically small and the results do not hold employing other robustness checks.

6.2. Bank capitalization

As predicted by H2, the loan rates applied by less capitalized IRB banks to public firms may be more sensitive to financial market conditions. Table 4 shows the results of Eq. (2).²⁹ The main findings discussed in Section 6.1 remain unchanged both for SA loans, column (1), and for IRB ones, column (2). Moreover, the coefficient of the interaction *Public*·*HighBankCapital* is not significant in both columns. This suggests that banks with high capital ratios do not evaluate public firms differently than other lenders.

The results in column (2) show that *Public·VSTOXX·HighBankCapital* has a significant negative impact on interest rates only for loans granted by IRB banks. An increase in VSTOXX of one standard deviation (8 units) leads to a reduction in the loan cost advantage of public firms of 35 bps for loans granted by IRB banks with lower capital ratios, and of about 1 bp for loans granted by IRB banks with a capital ratio above the 75th percentile of the distribution.³⁰ These estimates imply that the loan rates applied to public borrowers by latter IRB banks are almost no sensitive to financial market conditions. It should be noted that the significance of *Public·VSTOXX·HighBankCapital* is low, probably because of the limited variability of capital ratios across IRB banks and over time. However, keeping this caveat in mind, the results suggest that more capitalized IRB financial institutions are able to mitigate the reduction in the loan cost advantage of being public during high volatility periods.

By comparison, the estimates reported in column (1) show that the triple interaction between *Public*, *VSTOXX*, and *HighBankCapital* does not have a significant impact on the interest rates applied by SA banks. Overall, these findings support the hypothesis *H2*.

7. Robustness checks

The following sections present further robustness checks to confirm the validity of the main findings. The first test controls for the stock volatility of each public borrower. The second and the third sets of robustness checks focus on the heterogeneity of banks and borrowers, respectively. The fourth test presents an alternative sample restriction by adopting a propensity score matched sample.

The following robustness checks focus mostly on Eq. (1) (hypothesis H1). However, also the main findings related to hypothesis H2 hold when the same tests, unreported for space considerations, are replicated by employing Eq. (2).

²⁹ The reduction in the number of observations is due to some missing in the time series of consolidated bank capital ratios. ³⁰ Using the estimates in column (1): if *HighBankCapital*=0, $8 \cdot 0.044 = 0.352$; if *HighBankCapital*=1, $8 \cdot (0.044 - 0.043) =$

^{0.008.}

7.1. The stock volatility of public firms

Hypothesis *H1* suggests that IRB banks incorporate a change in financial market conditions in their assessment of the public firms' creditworthiness. If *H1* holds, the sensitivity to the stock volatility of each borrower should be greater for interest rates on IRB loans than for those on SA ones. Therefore, the interest rates applied to public firms with higher stock volatility by IRB banks should be significantly greater than those charged by SA intermediaries.

To verify this implication of H1, Eq. (1) is estimated by employing the sample of IRB and SA loans granted only to public firms and by introducing *FirmVol*, the stock volatility of firm *i* estimated as the standard deviation of stock returns over the year prior to *t*, and the interaction *FirmVol*·*IRB*.

Table 5 presents the results of this test. Since the coefficient of *FirmVol* is not significant when controlling for borrowers' characteristics, the estimates indicate that firm stock volatility does not affect the interest rate applied by SA banks. In contrast, the coefficient of *FirmVol·IRB* is significant and positive, implying that the interest rates applied to public firms with higher stock volatility by IRB banks are significantly greater than those charged by SA banks. These findings are consistent with *H1* and strongly suggest that the adoption of the IRB approach has made loan rates for public firms significantly more market-oriented.

7.2. Bank heterogeneity

Notwithstanding the adoption of the IRB method is a voluntary decision of banks, an IRB lender cannot evaluate the borrowers in the same portfolio with different methods after the validation of the supervisors. In addition, it cannot immediately switch its approach after (or before) a worsening of quarterly financial market conditions. These elements minimize endogeneity concerns related to the choice of the regulatory approach. However, since only few large banks have decided to adopt internal models in Italy, IRB lenders may be significantly different from SA financial institutions. In the baseline model, bank heterogeneity is controlled by including bank-time fixed effects. However, this section presents three additional robustness checks to verify whether bank heterogeneity may drive the main findings.

(*i*) In the first test, the banks that have extended only IRB or SA loans are excluded from the sample. Therefore, this robustness check examines the subsample of SA banks that have adopted the IRB approach during the examined period, focusing on the discontinuity observed after the transition from the standardized approach to the IRB method. By exploiting the variation within banks, it is verified whether the main results are robust to examining the same lender that has granted both IRB and SA loans over the considered period.

Column (1) of Table 6 shows the results of Eq. (1) estimated by employing this subsample of banks. The model controls for the change in the adopted approach for each bank by including *IRB* and interacting this variable with *Public* and *Public*·*VSTOXX*. The results of this test confirm the main findings. Since the coefficient of *Public* is not significant, the interest rates on SA loans applied to public companies are not significantly different from those charged to private firms. Also the impact of *Public*·*VSTOXX* on loan rates is not significant, implying that the interest rates on SA loans for public firms are not sensitive to the financial market climate.

In contrast, the significant negative coefficient of *Public*·*IRB* indicates that, after the adoption of the IRB approach, the loan rates applied to public firms by these banks are relatively lower than those charged to private companies. Finally, the positive coefficient of *Public*·*VSTOXX*·*IRB* suggests that, for IRB loans, the loan cost advantage of public firms significantly decreases after a worsening of the financial market climate. Therefore, these findings are consistent with *H1*.

(*ii*) The second test verifies whether the results are robust to excluding the loans granted by the two major Italian banking groups. Both banks have adopted the IRB approach during the examined period. Given their relevance, the loan pricing of these lenders may drive the findings reported in previous sections. Column (2) of Table 6 shows the results of Eq. (1) estimated by employing the sample of SA and IRB loans without those granted by the two major Italian banking groups. The estimates indicate that the coefficient of *Public·VSTOXX·IRB* remains significant, confirming that the main findings are not exclusively due to the loan pricing of the major IRB banks. However, the considerable contribution of the two major banking group is witnessed by the lower significance of the triple interaction coefficient compared with that observed in column (3) of Table 3, which includes the loans granted by these intermediaries.

(*iii*) The third robustness check is a falsification test to mitigate potential concerns about structural differences in loan pricing across banks that are unrelated to the adopted regulatory approach. In this test the sample consists of loans granted between 2008 and 2012 by banks adopting SA throughout the entire period.³¹ The model is estimated by replacing the variable *IRB* with *FutureIRB*, a dummy variable equal to 1 if the bank *j* has adopted the IRB approach after 2012 and 0 otherwise. Consequently, the IRB method is "imputed" to financial intermediaries that are still adopting the standardized approach in *t*.

Table 6 shows the results of Eq. (1) estimated by employing this sample and introducing the interactions *Public*·*FutureIRB* and *Public*·*VSTOXX*·*FutureIRB*. The coefficients of both variables are not significant, implying that the loan pricing in 2008-12 is not statistically different between the SA intermediaries that will change their approach after 2012 and the other SA banks. Therefore,

³¹ The results are robust to considering also other time windows.

this result suggests that the main findings are strongly related to the approach adopted by banks and are not driven by characteristics observable before the regulatory method shift.

7.3. Borrower heterogeneity

This section addresses two potential issues deriving from (*i*) a different sample composition between periods of low and high volatility and (*ii*) firm sorting across IRB and SA banks.

(*i*) The characteristics of borrowers during periods of low financial market volatility may be significantly different from those observed in high volatility times. After a financial shock, for instance, banks might not extend credit to highly risky borrowers. In this case, risky firms will not be included in the analyzed sample during high volatility periods. The different sample composition between low and high volatility periods may affect the main findings.

To address this potential issue, Eq. (1) is estimated by considering a constant sample of borrowers that have received at least one IRB loan in both low and high volatility periods. In this analysis, a quarter is defined as a "low volatility period" if the VSTOXX value in the quarter t-1 was lower than the 25th percentile of the distribution of the index over the examined period (2008-2018). In contrast, a quarter is defined as a "high volatility period" if the VSTOXX value in the quarter t-1 was greater than the 75th percentile of the distribution of the index over the analyzed period.

Table 7 reports the results of this test. First, the model in column (1) considers all IRB loans extended to the constant sample of borrowers in all quarters. Second, the analysis in column (2) is restricted to the IRB loans granted to the constant sample only in low and high volatility quarters. The coefficient of *Public·VSTOXX* remains significant and positive in both analyses, suggesting that the results are robust to considering a different sample composition between low and high volatility periods.

(*ii*) The second set of tests addresses the bias related to the potential borrower sorting across the two groups of lenders. Indeed, the firm decision to borrow prevalently from IRB or SA banks may be endogenous and may be due to several characteristics that could also affect loan rates.

The first test examines only companies borrowing prevalently from IRB banks (i.e. those that receive a share of IRB credit greater than the median). Therefore, the sample of this analysis includes firms with similar choices in terms of lender type.

In addition, as argued in the literature (Repullo and Suarez, 2004), low-risk firms may tend to borrow mostly from IRB banks, while riskier companies may principally rely on SA ones. Also this sorting may affect the main findings: loan rates for riskier borrowers may have a limited upward variability because they are closer to the maximum loan rate (i.e. the usury threshold rate). To take into account also this potential bias, the second exercise is restricted to firms borrowing prevalently from IRB banks and with a credit score greater than six (i.e. dummy *HighRisk* equal to 1).

Columns (1) and (2) of Table 8 show the results of both tests. Consistent with the main findings, the coefficient of *Public*·*VSTOXX* remains significant and positive by examining both companies borrowing prevalently from IRB banks (column 1) and riskier borrowers (column 2).

Finally, in order to tackle additional firm-related endogeneity bias, Eq. (1) is estimated by introducing firm-quarter fixed effects, which control for borrower-level unobserved heterogeneity in each quarter. In this model bank-quarter FE are excluded and the impact of the borrower control variables is absorbed by firm-quarter FE. The results reported in column (3) of Table 8 show that the coefficient of *Public·VSTOXX·IRB* is positive and significant, implying that the main findings are robust to employing this model specification.

7.4. Propensity score matching

As underlined in Section 5.1, public firms are significantly different from private companies. Observable and unobservable differences may affect the comparability between the loan rates applied to public and private borrowers. To address this issue, in the baseline model the sample of private firms is restricted to those with access to the bond market. However, this restriction excludes from the analysis the subsample of private firms without access to market financing, which is the most common type of borrower in the Italian economic system.

This section presents an alternative sample restriction that allows including also previously excluded firms. A technique based on the propensity score matching (Heckman et al., 1997; Rosenbaum and Rubin, 1983) is employed to compare listed companies and the overall sample of unlisted firms (private companies with or without access to the bond market). The adoption of this method allows examining firms that have a similar propensity to being public, mitigating potential endogeneity issues associated with the significant differences between public and private borrowers' characteristics.

First, the quarters in which the VSTOXX value is lower than the median over the examined period are separated from those in which the VSTOXX value is greater than the median. The firms that have not received at least one loan in both periods (i.e. when the VSTOXX index is above or below the median) are excluded from the sample.³² This allows examining a constant sample of borrowers.

³² This analysis does not adopt the 25th and 75th percentiles as the threshold values for not reducing excessively the number of available observations for the matching procedure.

In this analysis the treated units are companies listed on the stock market (public borrowers). In contrast, the control group consists of unlisted firms (private borrowers). The propensity score of each firm is estimated by using *Public* as the dependent variable and *Borrower Variables* as independent ones.³³ Afterward, the nearest-neighbor matching is applied. Each loan to a treated firm in a period in which the VSTOXX was below the median is matched with the nearest-neighbor loan, in terms of its propensity score, granted to a control company in the same period. Finally, loans granted to firms without a match are excluded. The balancing properties for the main variables of interest, presented in Table A.3 in the Appendix, suggest that the matching procedure has significantly reduced the differences in mean between public and private firms.

Table 9 reports the estimates of Eq. (1) obtained by adopting the matched sample of borrowers. Columns (1) and (2) show the results for the sample of loans granted by SA and IRB banks, respectively. The main findings remain unchanged also by adopting this sample. The interaction variable *Public*·*VSTOXX* does not have a significant impact on loan rates charged by SA banks. In contrast, for loans granted by IRB banks, an increase in financial market volatility leads to a rise in interest rates significantly greater for treated public firms than for control private companies.

8. Conclusions

This paper examines the impact of bank capital regulation on the loan cost difference between public and private firms. The main findings indicate that the adoption of the IRB approach has made loan rates significantly more market-oriented for public firms. For credit granted by IRB banks, public firms benefit from a significant loan cost advantage with respect to private companies when the level of financial instability is low. However, a worsening of financial market conditions reduces the cost advantage of being public, leading to a rise in interest rates on IRB loans greater for public firms than for private ones. The analysis documents that the negative impact of financial instability on the loan cost advantage of public borrowers is significantly greater if the IRB bank has a lower capital ratio, suggesting that the main effect described in the analysis is mostly determined by banks that kept smaller capital buffers over the regulatory minimum.

In contrast, for credit granted by SA banks, public borrowers do not benefit from a significant loan cost advantage compared with private ones and a change in financial market conditions has a statistically similar effect on the borrowing costs of both types of companies.

The results are robust to considering the heterogeneity of banks and borrowers and using a matched sample.

³³ This model does not include *IntCovRatio* in the set of independent variables because the coefficient of this variable is not significant in the propensity score equation.

A caveat of this analysis is that the sample includes a limited number of borrowers. This restriction is common in the literature on public companies in European countries because only a small fraction of the firm population is listed on a stock exchange.

With due caution given the sample characteristics, the upside of these findings is that a marketoriented pricing of loans may lead to a more efficient allocation of resources in the financial system (Norden and Wagner, 2008). On the other hand, the main risk associated with market-based loan rates is that adverse shocks to financial markets may induce more volatility into public firms' borrowing costs, increasing the cost of bank credit and, in turn, worsening their financial conditions (Ivanov et al., 2016). Increasing dependence of loan rates for public borrowers on financial market conditions may reduce the stabilizing influence over the business cycle of bank credit, which is generally less sensitive to macroeconomic conditions than market financing (Norden and Wagner, 2008). As a result, this effect may reduce the net benefits of listing, decreasing the propensity of going or remaining public as predicted by the model of Doidge et al. (2017).

Given the potential implications associated with the adoption of the IRB approach, the results of this work support the orientation of regulators to enhance the scrutiny of internal models (BCBS, 2017).³⁴ Finally, the analysis highlights that bank capital plays a crucial role in reducing the transmission of unexpected shocks to borrowers. Indeed, large capital buffers allow IRB banks to offset almost completely the volatility in loan rates for public borrowers during uncertain financial market conditions.

³⁴ The reform constrains the use of internal models for large corporates and it also introduces "input floor" values for bank-estimated IRB parameters. The new framework will be implemented on 1 January 2023.

Appendix

Table A.1

Variables description.

Variable	Description	Source
Dependent Variable		
Rate	Interest rate on revolving credit lines granted by bank <i>j</i> to firm <i>i</i> on quarter <i>t</i> .	CR
Key Explanatory Varia	bles	
Public	Dummy variable equal to 1 if the firm <i>i</i> is listed in <i>t</i> and 0 otherwise.	Cerved
VSTOXX	Value of the VSTOXX index at the end of the quarter <i>t</i> -1.	Datastream
HighBankCapital	Dummy variable equal to 1 if the bank j has a capital ratio above the 75th percentile of the distribution of capital ratios in the year preceding t and 0 otherwise.	Supervisory Reports
IRB	Dummy variable equal to 1 if the bank j adopts the IRB approach in t and 0 otherwise.	Supervisory and Pillar 3 information
FirmVol	The stock volatility of firm <i>i</i> estimated as the standard deviation of stock returns over the year prior to <i>t</i> .	Datastream
FutureIRB	Dummy variable equal to 1 if the bank j has adopted the IRB approach after 2012 and 0 otherwise.	Supervisory and Pillar 3 information
Relationship Variables		
Share	The share of total credit (revolving credit lines, term loans, loans backed by accounts receivables) granted by bank <i>j</i> to firm <i>i</i> in <i>t</i> .	CR
RelDuration	Dummy equal to 1 if the bank-firm relationship started at least in the year before t .	CR
Borrower Variables		
Size	Logarithm of the total assets of firm <i>i</i> in the year preceding <i>t</i> .	Cerved
HighRisk	Dummy variable equal to 1 if the credit score assigned by Cerved to firm i in the year preceding t is greater than six and 0 otherwise.	Cerved
Leverage	Ratio of total debt divided by the book value of assets of the firm i in the year preceding t .	Cerved
Fixed Assets	Ratio of fixed assets to total assets of the firm i in the year preceding t .	Cerved
EBITDA	Ratio of EBITDA to total assets of the firm <i>i</i> in the year preceding <i>t</i> .	Cerved
IntCovRatio	Ratio of EBITDA to interest expense of the firm i in the year preceding t .	Cerved
Industry Dummies	Indicator variables for the industry of firm <i>i</i> based on 2-digit ATECO.	Cerved

Table A.2

Balancing test: pre and post-matching *t*-test differences between public and private firms.

	Pre-matching	Post-matching
Size	3.17***	0.07
	(0.00)	(0.18)
HighRisk	-0.04***	0.02
	(0.00)	(0.20)
Leverage	-0.11***	-0.01*
	(0.00)	(0.06)
Fixed Assets	-0.12***	-0.01
	(0.00)	(0.90)
EBITDA	-0.02***	-0.01
	(0.00)	(0.59)

Differences in mean with respect to the group of control companies. ***, **, and * denote significance at the 1%, 5%, and 10% level in a *t*-test for means (*p*-values in parentheses).

References

- Accornero, M., Finaldi Russo, P., Guazzarotti, G., Nigro, V., 2018. Missing investors in the Italian corporate bond market, Questioni di Economia e Finanza (Occasional Papers), Bank of Italy.
- Acharya, V. V., Almeida, H., Ippolito, F., Perez, A., 2014. Credit lines as monitored liquidity insurance: Theory and evidence. Journal of Financial Economics 112, 287–319.
- Albertazzi, U., Marchetti, D.J., 2010. Credit Supply, Flight to Quality and Evergreening: An Analysis of Bank-Firm Relationships after Lehman, Temi di Discussione (Working Papers), Bank of Italy.
- Altman, E.I., Marco, G., Varetto, F., 1994. Corporate distress diagnosis: Comparisons using linear discriminant analysis and neural networks (the Italian experience). Journal of Banking and Finance 18, 505–529.
- Barclays, 2015. Pillar 3 Report Return to stability.
- BCBS, 2017. High-level summary of Basel III reforms.
- Behn, M., Haselmann, R., Vig, V., 2016a. The limits of model-based regulation, ECB Working Paper Series.
- Behn, M., Haselmann, R., Wachtel, P., 2016b. Procyclical Capital Regulation and Lending. Journal of Finance 71, 919–956.
- Berg, T., Koziol, P., 2017. An analysis of the consistency of banks' internal ratings. Journal of Banking & Finance 78, 27–41.
- Berger, A.N., Udell, G.F., 1995. Relationship Lending and Lines of Credit in Small Firm Finance. The Journal of Business 68, 351–379.
- Bharath, S.T., Dahiya, S., Saunders, A., Srinivasan, A., 2011. Lending relationships and loan contract terms. Review of Financial Studies 24, 1141–1203.
- Bharath, S.T., Dittmar, A.K., 2010. Why do firms use private equity to opt out of public markets? Review of Financial Studies 23, 1771–1818.
- Boot, A.W.A., 2000. Relationship Banking: What Do We Know? Journal of Financial Intermediation 9, 7–25.
- Brav, O., 2009. Access to capital, capital structure, and the funding of the firm. Journal of Finance 64, 263–308.
- Cannata, F., Casellina, S., Guidi, G., 2012. Inside the Labyrinth of Basel Risk-Weighted Assets: How Not to Get Lost, Questioni di Economia e Finanza (Occasional Papers), Bank of Italy.
- Cucinelli, D., Battista, M.L. Di, Marchese, M., Nieri, L., 2018. Credit risk in European banks: The bright side of the internal ratings based approach. Journal of Banking and Finance 93, 213–229.

- Degryse, H., Ongena, S., 2005. Distance, lending relationships, and competition. Journal of Finance 231–266.
- Denis, D.J., Mihov, V.T., 2003. The choice among bank debt, non-bank private debt, and public debt: Evidence from new corporate borrowings. Journal of Financial Economics 70, 3–28.
- Doidge, C., Karolyi, G.A., Stulz, R.M., 2017. The U.S. listing gap. Journal of Financial Economics 123, 464–487.
- EBA, 2013. Report on the pro-cyclicality of capital requirements under the Internal Ratings Based Approach.
- Elsas, R., 2005. Empirical determinants of relationship lending. Journal of Financial Intermediation 14, 32–57.
- Firestone, S., Rezende, M., 2016. Are Banks' Internal Risk Parameters Consistent? Evidence from Syndicated Loans. Journal of Financial Services Research 50, 211–242.
- Gallo, R., 2019. The loan cost advantage of public firms and financial market conditions: evidence from the European syndicated loan market, Temi di Discussione (Working Papers), Bank of Italy.
- Hale, G., Santos, J.A.C., 2008. The decision to first enter the public bond market: The role of firm reputation, funding choices, and bank relationships. Journal of Banking and Finance 32, 1928– 1940.
- Heckman, J.J., Ichimura, H., Todd, P.E., 1997. Matching Evidence Job As An Econometric Estimator: Evidence from Evaluating a Job Training Programme. Review of Economic Studies 64, 605–654.
- Ioannidou, V., Ongena, S., 2010. "Time for a change": Loan conditions and bank behavior when firms switch banks. Journal of Finance 65, 1847–1877.
- Ivanov, I.T., Santos, J.A.C., Vo, T., 2016. The transformation of banking: Tying loan interest rates to borrowers' CDS spreads. Journal of Corporate Finance 38, 150–165.
- Kashyap, A., Stein, J., 2004. Cyclical implications of the Basel II capital standards. Federal Reserve Bank of Chicago 28, 18–31.
- Le Leslé, V., Avramova, S., 2012. Revisiting Risk-Weighted Assets "Why Do RWAs Differ Across Countries and What Can Be Done About It?," IMF Working Paper.
- Leuz, C., Triantis, A., Yue Wang, T., 2008. Why do firms go dark? Causes and economic consequences of voluntary SEC deregistrations. Journal of Accounting and Economics 45, 181–208.
- Mariathasan, M., Merrouche, O., 2014. The manipulation of basel risk-weights. Journal of Financial Intermediation 23, 300–321.

- Marosi, A., Massoud, N., 2007. Why Do Firms Go Dark? The Journal of Financial and Quantitative Analysis 42, 421–442.
- Norden, L., Wagner, W., 2008. Credit derivatives and loan pricing. Journal of Banking and Finance 32, 2560–2569.
- Pagano, M., Panetta, F., Zingales, L., 1998. Why do companies go public? An empirical analysis. The Journal of Finance 53, 27–64.
- Pérez Montes, C., Trucharte Artigas, C., Cristófoli, M.E., Lavín San Segundo, N., 2018. The impact of the IRB approach on the risk weights of European banks. Journal of Financial Stability 39, 147–166.
- Plosser, M.C., Santos, J.A.C., 2014. Banks' incentives and the quality of internal risk models, Federal Reserve Bank of New York Staff Reports.
- Rajan, R.G., 1992. Insiders and Outsiders: The Choice between Informed and Arm's Length Debt. The Journal of Finance 47, 1367–1400.
- Repullo, R., Suarez, J., 2004. Loan pricing under Basel capital requirements. Journal of Financial Intermediation 13, 496–521.
- Repullo, R., Suarez, J., 2013. The procyclical effects of bank capital regulation. Review of Financial Studies 26, 452–490.
- Rosenbaum, P.R., Rubin, D.B., 1983. The Central Role of the Propensity Score in Observational Studies for Causal Effects. Biometrika 70, 41–55.
- Santos, J.A.C., Winton, A., 2008. Bank loans, bonds, and information monopolies across the business cycle. Journal of Finance 63, 1315–1359.
- Saunders, A., Steffen, S., 2011. The costs of being private: Evidence from the loan market. Review of Financial Studies 24, 4091–4122.
- Sette, E., Gobbi, G., 2015. Relationship lending during a financial crisis. Journal of the European Economic Association 13, 453–481.

Tables and figures

Figure 1

The quarterly time series of the VSTOXX index between 2008Q1 and 2018Q2.



Bank-firm relationships included in the sample.

	Public Firms	Private Firms
Share of relationships with IRB banks	41.47%	44.88%
Share of credit granted by IRB banks	57.06%	52.28%

Table 2

Summary statistics comparing public and private firms.

Variable		Public Firn	ıs]	Private Firr	ns	Differe	ences
	Mean	Median	Std. Dev.	Mean	Median	Std. Dev.	Diff. in mean ¹	Diff. in median ²
Rate (%)	8.44	7.48	4.02	8.85	8.23	3.76	-0.41***	-0.74***
VSTOXX	26.23	24.06	8.33	25.70	23.38	8.29	0.54^{***}	0.67^{***}
HighBankCapital	0.23	0.00	0.42	0.23	0.00	0.42	0.01	0.00
Share	0.14	0.09	0.15	0.16	0.11	0.15	-0.02***	-0.02***
RelDuration	0.91	1.00	0.29	0.93	1.00	0.25	-0.03***	0.00^{***}
Size (<i>ln</i>)	12.02	11.89	1.89	10.57	10.52	1.08	1.45***	1.37***
HighRisk	0.63	1.00	0.48	0.59	1.00	0.49	0.03***	0.00^{***}
Leverage	0.70	0.70	0.23	0.77	0.77	0.23	-0.06***	-0.07***
FixedAsset	0.13	0.07	0.16	0.28	0.22	0.25	-0.15***	-0.15***
EBITDA	0.02	0.02	0.08	0.03	0.04	0.07	-0.01***	-0.01***
IntCovRat	1.87	1.22	9.04	2.06	1.77	5.99	-0.20**	-0.55***

¹ *** Significant at 1%, ** significant at 5%, * significant at 10% in a *t*-test for means. ² *** Significant at 1%, ** significant at 5%, * significant at 10% in a Pearson χ^2 test for medians.

The interest rates applied to public and private firms by SA and IRB banks.

	(1)	(2)	(3)
	SA banks	IRB banks	All banks
Public	0.203	-0.731***	0.291
	(0.397)	(0.005)	(0.211)
Public·VSTOXX	0.004	0.039***	0.004
	(0.655)	(0.000)	(0.607)
Public·IRB	-	-	-1.099***
			(0.001)
Public·VSTOXX·IRB	-	-	0.034***
			(0.007)
Share	-1.940***	-1.007***	-1.494***
	(0.000)	(0.000)	(0.000)
RelDuration	0.347**	0.527**	0.412***
	(0.016)	(0.031)	(0.001)
Size	-0.503***	-0.642***	-0.563***
	(0.000)	(0.000)	(0.000)
HighRisk	0.815***	0.976***	0.898***
	(0.000)	(0.000)	(0.000)
Leverage	0.664***	0.138	0.402***
	(0.000)	(0.409)	(0.001)
FixedAsset	-0.263*	-0.291*	-0.271***
	(0.060)	(0.058)	(0.009)
EBITDA	-2.521***	-1.402**	-1.986***
	(0.000)	(0.011)	(0.000)
IntCovRat	0.009	-0.003	0.003
	(0.142)	(0.544)	(0.378)
Industry dummies	Yes	Yes	Yes
Bank-quarter FE	Yes	Yes	Yes
Observations	16056	13850	29906
Adj R-squared	0.167	0.149	0.160

The table shows the results obtained from the estimation of Eq. (1) by analyzing loans granted by SA banks, in column (1), and those granted by IRB banks, in column (2). Column (3) presents the results of Eq. (1) estimated by employing the full sample of loans and by introducing the interaction variables *Public*·*IRB* and *Public*·*VSTOXX*·*IRB*. The dependent variable is *Rate*, the interest rate on revolving credit lines granted by bank j to firm i on quarter t. Robust p-values in parentheses. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

The effect of bank capitalization on the interest rates applied to public and private firms by SA and IRB banks.

	(1)	(2)
	SA banks	IRB banks
Public	-0.071	-0.765**
	(0.812)	(0.023)
Public·VSTOXX	0.012	0.044^{***}
	(0.286)	(0.000)
Public·HighBankCapital	1.155	0.650
	(0.175)	(0.307)
Public·VSTOXX·HighBankCapital	-0.026	-0.043*
	(0.324)	(0.094)
Share	-1.799***	-1.033***
	(0.000)	(0.000)
RelDuration	0.250	0.426
	(0.138)	(0.115)
Size	-0.507***	-0.682***
	(0.000)	(0.000)
HighRisk	0.763***	0.971***
	(0.000)	(0.000)
Leverage	0.641***	0.095
	(0.000)	(0.593)
FixedAsset	-0.058	-0.241
	(0.709)	(0.142)
EBITDA	-2.720***	-1.262**
	(0.000)	(0.045)
IntCovRat	0.016	0.002
	(0.152)	(0.740)
Industry dummies	Yes	Yes
Bank-quarter FE	Yes	Yes
Observations	13163	12026
Adj R-squared	0.157	0.147

The table shows the results obtained from the estimation of Eq. (2) by analyzing loans granted by SA banks, in column (1), and those granted by IRB banks, in column (2). The dependent variable is *Rate*, the interest rate on revolving credit lines granted by bank j to firm i on quarter t. Robust p-values in parentheses. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

The interest rates applied to public firms on SA and IRB loans by considering firm stock volatility.

	(1)
	Public firms
FirmVol	-0.306
	(0.283)
FirmVol·IRB	0.823**
	(0.035)
Share	-2.158***
	(0.000)
RelDuration	0.218
	(0.404)
Size	-0.664***
	(0.000)
HighRisk	0.135
Ŧ	(0.317)
Leverage	0.431
E: 14 ((0.101)
FixedAsset	-1.562***
EBITDA	(0.000) -1.206
EBIIDA	(0.183)
IntCovRat	0.006
Inteovicat	(0.340)
	· · · ·
Industry dummies	Yes
Bank-quarter FE	Yes
Observations	5747
Adj R-squared	0.202

The table presents the results of Eq. (1) estimated by employing a sample of IRB and SA loans granted only to public firms and introducing *FirmVol* and the interaction *FirmVol·IRB*. The dependent variable is *Rate*, the interest rate on revolving credit lines granted by bank *j* to firm *i* on quarter *t*. Robust *p*-values in parentheses. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

The interest rates applied to public and private firms on SA and IRB loans by considering bank heterogeneity.

	(1)	(2)	(3)
	Subsample of banks	Excl. major bank. groups	Falsification test
Public	0.362	0.197	0.702
	(0.341)	(0.408)	(0.105)
Public·VSTOXX	-0.009	0.005	-0.002
	(0.492)	(0.571)	(0.905)
Public·IRB	-1.486***	-1.070**	-
	(0.002)	(0.015)	
Public·VSTOXX·IRB	0.054***	0.031*	-
	(0.001)	(0.060)	
Public·FutureIRB	-	-	-0.255
			(0.667)
Public·VSTOXX·FutureIRB	-	-	0.002
			(0.924)
Share	-1.278***	-1.759***	-1.813***
	(0.000)	(0.000)	(0.000)
RelDuration	0.392*	0.389***	0.187
	(0.055)	(0.003)	(0.307)
Size	-0.614***	-0.544***	-0.566***
	(0.000)	(0.000)	(0.000)
HighRisk	0.978***	0.866***	0.990***
	(0.000)	(0.000)	(0.000)
Leverage	0.307*	0.277**	0.719***
	(0.051)	(0.042)	(0.008)
FixedAsset	-0.779***	-0.401***	-0.242
	(0.000)	(0.001)	(0.224)
EBITDA	-1.116**	-2.445***	-3.910***
	(0.028)	(0.000)	(0.000)
IntCovRat	-0.002	0.012*	0.056***
	(0.666)	(0.055)	(0.000)
Industry dummies	Yes	Yes	Yes
Bank-quarter FE	Yes	Yes	Yes
Observations	14984	22314	8928
Adj R-squared	0.153	0.157	0.182

Column (1) presents the results of Eq. (1) estimated by employing SA and IRB loans granted by a subsample of banks and by introducing the interaction variables *Public*·*IRB* and *Public*·*VSTOXX*·*IRB*. Column (2) presents the results of Eq. (1) estimated by employing the sample of SA and IRB loans without those granted by the two major Italian banking groups and by introducing the interaction variables *Public*·*IRB* and *Public*·*VSTOXX*·*IRB*. Column (3) shows the results of Eq. (1) estimated by employing loans granted between 2008 and 2012 by banks adopting SA throughout the entire period. The model includes the interaction variables *Public*·*VSTOXX*·*FutureIRB*. The dependent variable is *Rate*, the interest rate on revolving credit lines granted by bank *j* to firm *i* on quarter *t*. Robust *p*-values in parentheses. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

	(1) (2		
	Constant sample	Low and high vol. periods	
Public	-1.072***	-1.366***	
	(0.000)	(0.001)	
Public VSTOXX	0.052***	0.060***	
	(0.000)	(0.000)	
Share	-1.268***	-1.289***	
	(0.000)	(0.000)	
RelDuration	1.046***	1.130***	
	(0.000)	(0.008)	
Size	-0.703***	-0.711***	
	(0.000)	(0.000)	
HighRisk	0.874***	0.830***	
	(0.000)	(0.000)	
Leverage	-0.219	-0.093	
	(0.208)	(0.701)	
FixedAsset	-0.466***	-0.356	
	(0.003)	(0.158)	
EBITDA	-1.020	-0.679	
	(0.103)	(0.494)	
IntCovRat	-0.001	0.003	
	(0.890)	(0.791)	
Industry dummies	Yes	Yes	
Bank-quarter FE	Yes	Yes	
Observations	12151	5112	
Adj R-squared	0.161	0.172	

The table shows the results obtained from the estimation of Eq. (1) by considering a constant sample of borrowers that have received at least one IRB loan in both low and high volatility periods. In column (1) the model is estimated by considering all IRB loans extended to the constant sample of borrowers. In column (2) the model is estimated by considering only IRB loans granted in low and high volatility periods. The dependent variable is *Rate*, the interest rate on revolving credit lines granted by bank *j* to firm *i* on quarter *t*. Robust *p*-values in parentheses. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

The interest rates applied to public and private firms by considering borrower heterogeneity.

	(1)	(2)	(3)
	Prev. IRB Borr.	Prev. IRB HR Borr.	Firm-quarter FE
Public	-0.945***	-1.930***	-
	(0.005)	(0.000)	
Public·VSTOXX	0.052***	0.065***	-
	(0.000)	(0.000)	
IRB	-	-	1.780***
			(0.000)
IRB·VSTOXX	-	-	-0.039***
			(0.000)
Public·IRB	-	-	-1.098***
D 11' VETOXX IDD			(0.001)
Public·VSTOXX·IRB	-	-	0.030***
Share	-1.117***	-1.041***	(0.008) -0.655***
Share	(0.000)	(0.004)	(0.000)
RelDuration	1.087***	0.729*	0.473***
ReiDuration	(0.000)	(0.089)	(0.000)
Size	-0.637***	-0.733***	(0.000)
	(0.000)	(0.000)	
HighRisk	1.009***	-	-
6	(0.000)		
Leverage	-0.065	-0.366	-
C C	(0.746)	(0.137)	
FixedAsset	0.345	-0.036	-
	(0.109)	(0.906)	
EBITDA	-0.082	-3.417**	-
	(0.913)	(0.011)	
IntCovRat	-0.001	0.056**	-
	(0.931)	(0.011)	
Industry dummies	Yes	Yes	No
Bank-quarter FE	Yes	Yes	No
Firm-quarter FE	No	No	Yes
Observations	8150	4778	29906
Adj R-squared	0.138	0.155	0.343

Column (1) shows the results obtained from the estimation of Eq. (1) by considering only IRB loans granted to borrowers that receive a share of credit from IRB banks greater than the median. The estimates in column (2) take into account only IRB loans granted to high-risk borrowers (i.e. a credit score greater than six) that receive a share of credit from IRB financial institutions greater than the median. Column (3) shows the results of Eq. (1) estimated by introducing firm-quarter fixed effects. The dependent variable is *Rate*, the interest rate on revolving credit lines granted by bank *j* to firm *i* on quarter *t*. Robust *p*-values in parentheses. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

The interest rates applied to a matched sample of public and private firms.

	(1)	(2)
	SA banks	IRB banks
Public	-0.094	-1.282***
	(0.722)	(0.000)
Public·VSTOXX	0.006	0.047***
	(0.521)	(0.000)
Share	-2.273***	-1.011***
	(0.000)	(0.000)
RelDuration	0.183	0.163
	(0.142)	(0.488)
Size	-0.601***	-0.647***
	(0.000)	(0.000)
HighRisk	0.925***	0.985***
	(0.000)	(0.000)
Leverage	0.208	0.113
	(0.225)	(0.525)
FixedAsset	-1.236***	-1.142***
	(0.000)	(0.000)
EBITDA	-0.819***	-0.502**
	(0.000)	(0.014)
IntCovRat	-0.001*	-0.001
	(0.077)	(0.932)
Industry dummies	Yes	Yes
Bank-quarter FE	Yes	Yes
Observations	17415	13197
Adj R-squared	0.153	0.129

The table shows the results obtained from the estimation of Eq. (1) by adopting the matched sample of loans granted by SA banks, in column (1), and those granted by IRB banks, in column (2). The dependent variable is *Rate*, the interest rate on revolving credit lines granted by bank *j* to firm *i* on quarter *t*. Robust *p*-values in parentheses. ***, **, and * denote significance at the 1%, 5%, and 10% levels, respectively.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1264 *The impact of TLTRO2 on the Italian credit market: some econometric evidence*, by Lucia Esposito, Davide Fantino and Yeji Sung (February 2020).
- N. 1265 *Public credit guarantee and financial additionalities across SME risk classes*, by Emanuele Ciani, Marco Gallo and Zeno Rotondi (February 2020).
- N. 1266 Determinants of the credit cycle: a flow analysis of the extensive margin, by Vincenzo Cuciniello and Nicola di Iasio (March 2020).
- N. 1267 *Housing supply elasticity and growth: evidence from Italian cities*, by Antonio Accetturo, Andrea Lamorgese, Sauro Mocetti and Dario Pellegrino (March 2020).
- N. 1268 Public debt expansions and the dynamics of the household borrowing constraint, by António Antunes and Valerio Ercolani (March 2020).
- N. 1269 *Expansionary yet different: credit supply and real effects of negative interest rate policy*, by Margherita Bottero and Enrico Sette (March 2020).
- N. 1270 Asymmetry in the conditional distribution of euro-area inflation, by Alex Tagliabracci (March 2020).
- N. 1271 An analysis of sovereign credit risk premia in the euro area: are they explained by local or global factors?, by Sara Cecchetti (March 2020).
- N. 1272 *Mutual funds' performance: the role of distribution networks and bank affiliation*, by Giorgio Albareto, Andrea Cardillo, Andrea Hamaui and Giuseppe Marinelli (April 2020).
- N. 1273 Immigration and the fear of unemployment: evidence from individual perceptions in Italy, by Eleonora Porreca and Alfonso Rosolia (April 2020).
- N. 1274 Bridge Proxy-SVAR: estimating the macroeconomic effects of shocks identified at high-frequency, by Andrea Gazzani and Alejandro Vicondoa (April 2020).
- N. 1275 *Monetary policy gradualism and the nonlinear effects of monetary shocks*, by Luca Metelli, Filippo Natoli and Luca Rossi (April 2020).
- N. 1276 Spend today or spend tomorrow? The role of inflation expectations in consumer behaviour, by Concetta Rondinelli and Roberta Zizza (April 2020).
- N. 1277 *Going the extra mile: effort by workers and job-seekers*, by Matthias S. Hertweck, Vivien Lewis and Stefania Villa (June 2020).
- N. 1278 *Trainspotting: board appointments in private firms*, by Audinga Baltrunaite and Egle Karmaziene (June 2020).
- N. 1279 *The role of bank supply in the Italian credit market: evidence from a new regional survey*, by Andrea Orame (June 2020).
- N. 1280 The non-linear effects of the Fed asset purchases, by Alessio Anzuini (June 2020).
- N. 1281 *The effects of shop opening hours deregulation: evidence from Italy*, by Lucia Rizzica, Giacomo Roma and Gabriele Rovigatti (June 2020).
- N. 1282 *How do house prices respond to mortgage supply*?, by Guglielmo Barone, Francesco David, Guido de Blasio and Sauro Mocetti (June 2020).
- N. 1283 *The macroeconomics of hedging income shares*, by Adriana Grasso, Juan Passadore and Facundo Piguillem (June 2020).
- N. 1284 Uncertainty matters: evidence from a high-frequency identification strategy, by Piergiorgio Alessandri, Andrea Gazzani and Alejandro Vicondoa (June 2020).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2018

- ACCETTURO A., V. DI GIACINTO, G. MICUCCI and M. PAGNINI, Geography, productivity and trade: does selection explain why some locations are more productive than others?, Journal of Regional Science, v. 58, 5, pp. 949-979, WP 910 (April 2013).
- ADAMOPOULOU A. and E. KAYA, *Young adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, WP 1038 (November 2015).
- ANDINI M., E. CIANI, G. DE BLASIO, A. D'IGNAZIO and V. SILVESTRINI, *Targeting with machine learning:* an application to a tax rebate program in Italy, Journal of Economic Behavior & Organization, v. 156, pp. 86-102, WP 1158 (December 2017).
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces*, Regional Science and Urban Economics, v. 70, pp. 352-59, **WP 1057 (March 2016)**.
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, v.31, 6, pp. 2113-2156, WP 1097 (February 2017).
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, WP 909 (April 2013).
- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. Husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, WP 1063 (April 2016).
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME borrowing*. A RDD evaluation, Journal of Banking & Finance, v. 96, pp. 73-86, WP 1111 (April 2017).
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010 (June 2015).**
- LIBERATI D., An estimated DSGE model with search and matching frictions in the credit market, International Journal of Monetary Economics and Finance (IJMEF), v. 11, 6, pp. 567-617, WP 986 (November 2014).
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, v. 14, 1, pp. 35-71, WP 1025 (July 2015).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- RIGON M. and F. ZANETTI, *Optimal monetary policy and fiscal policy interaction in a non_ricardian economy,* International Journal of Central Banking, v. 14 3, pp. 389-436, WP 1155 (December 2017).

SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

2019

- ALBANESE G., M. CIOFFI and P. TOMMASINO, *Legislators' behaviour and electoral rules: evidence from an Italian reform*, European Journal of Political Economy, v. 59, pp. 423-444, **WP 1135 (September 2017).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, v. 15, 4, pp. 55-80, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis, Italian Economic Journal / Rivista italiana degli economisti, v. 5, 1, pp. 17-37, WP 1086 (October 2016).
- BASSO G., F. D'AMURI and G. PERI, *Immigrants, labor market dynamics and adjustment to shocks in the euro area,* IMF Economic Review, v. 67, 3, pp. 528-572, WP 1195 (November 2018).
- BATINI N., G. MELINA and S. VILLA, *Fiscal buffers, private debt, and recession: the good, the bad and the ugly,* Journal of Macroeconomics, v. 62, WP 1186 (July 2018).
- BURLON L., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of an open-ended asset purchase programme*, Journal of Policy Modeling, v. 41, 6, pp. 1144-1159, **WP 1185 (July 2018).**
- BUSETTI F. and M. CAIVANO, Low frequency drivers of the real interest rate: empirical evidence for advanced economies, International Finance, v. 22, 2, pp. 171-185, WP 1132 (September 2017).
- CAPPELLETTI G., G. GUAZZAROTTI and P. TOMMASINO, *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, Fiscal Studies, v. 40, 2, pp. 211-237, **WP 938 (November 2013).**
- CARDANI R., A. PACCAGNINI and S. VILLA, Forecasting with instabilities: an application to DSGE models with financial frictions, Journal of Macroeconomics, v. 61, WP 1234 (September 2019).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, Fiscal consolidation by intergovernmental transfers cuts? The unpleasant effect on expenditure arrears, Economic Modelling, v. 77, pp. 266-275, WP 985 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, Regional Science and Urban Economics, v. 75, pp. 1-21, WP 1112 (April 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, WP 985 (November 2014).
- CIAPANNA E. and M. TABOGA, *Bayesian analysis of coefficient instability in dynamic regressions*, Econometrics, MDPI, Open Access Journal, v. 7, 3, pp.1-32, WP 836 (November 2011).
- COLETTA M., R. DE BONIS and S. PIERMATTEI, *Household debt in OECD countries: the role of supply-side* and demand-side factors, Social Indicators Research, v. 143, 3, pp. 1185–1217, **WP 989 (November** 2014).
- COVA P., P. PAGANO and M. PISANI, *Domestic and international effects of the Eurosystem Expanded Asset Purchase Programme*, IMF Economic Review, v. 67, 2, pp. 315-348, **WP 1036 (October 2015).**
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, v. 23, 8. pp. 3457-2482, **WP 1174 (April 2018).**
- FERRERO G., M. GROSS and S. NERI, *On secular stagnation and low interest rates: demography matters,* International Finance, v. 22, 3, pp. 262-278, **WP 1137 (September 2017).**
- FOA G., L. GAMBACORTA, L. GUISO and P. E. MISTRULLI, *The supply side of household finance*, Review of Financial Studies, v.32, 10, pp. 3762-3798, **WP 1044 (November 2015).**
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, v. 78, pp. 118-133, WP 1167 (March 2018).
- GOMELLINI M., D. PELLEGRINO and F. GIFFONI, *Human capital and urban growth in Italy*,1981-2001, Review of Urban & Regional Development Studies, v. 31, 2, pp. 77-101, **WP 1127 (July 2017).**
- MAGRI S., Are lenders using risk-based pricing in the Italian consumer loan market? The effect of the 2008 crisis, Journal of Credit Risk, v. 15, 1, pp. 27-65, WP 1164 (January 2018).
- MAKINEN T., A. MERCATANTI and A. SILVESTRINI, *The role of financial factors for european corporate investment*, Journal of International Money and Finance, v. 96, pp. 246-258, **WP 1148 (October 2017).**
- MIGLIETTA A., C. PICILLO and M. PIETRUNTI, *The impact of margin policies on the Italian repo market*, The North American Journal of Economics and Finance, v. 50, **WP 1028 (October 2015).**

- MONTEFORTE L. and V. RAPONI, Short-term forecasts of economic activity: are fortnightly factors useful?, Journal of Forecasting, v. 38, 3, pp. 207-221, WP 1177 (June 2018).
- NERI S. and A. NOTARPIETRO, Collateral constraints, the zero lower bound, and the debt-deflation mechanism, Economics Letters, v. 174, pp. 144-148, WP 1040 (November 2015).
- PEREDA FERNANDEZ S., *Teachers and cheaters. Just an anagram?*, Journal of Human Capital, v. 13, 4, pp. 635-669, WP 1047 (January 2016).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, v. 23, 2, pp. 590-624, WP 871 (July 2012).

2020

- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, WP 1110 (April 2017).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, **WP 1059 (March 2016).**
- RIZZICA L., Raising aspirations and higher education. Evidence from the UK's widening participation policy, Journal of Labor Economics, v. 38, 1, pp. 183-214, WP 1188 (September 2018).

FORTHCOMING

- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, **WP 974 (October 2014).**
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, *Discretion and supplier selection in public procurement*, Journal of Law, Economics, and Organization, WP 1178 (June 2018).
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, WP 1201 (November 2018).
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy,* Journal of International Economics, **WP 1032 (October 2015).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome*, Journal of Regional Science, WP 1208 (February 2019).
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, **WP 1152 (December 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, WP 1156 (December 2017).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises*, Review of World Economics, WP 1217 (April 2019).
- GERALI A. and S. NERI, *Natural rates across the Atlantic*, Journal of Macroeconomics, WP 1140 (September 2017).
- LIBERATI D. and M. LOBERTO, *Taxation and housing markets with search frictions*, Journal of Housing Economics, WP 1105 (March 2017).
- LOSCHIAVO D., Household debt and income inequality: evidence from Italian survey data, Review of Income and Wealth, WP 1095 (January 2017).
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility*, Journal of Human Resources, WP 1182 (July 2018).

- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, **WP 1200** (November 2018).
- PANCRAZI R. and M. PIETRUNTI, *Natural expectations and home equity extraction*, Journal of Housing Economics, WP 984 (November 2014).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, WP 1092 (January 2017).

RAINONE E., The network nature of otc interest rates, Journal of Financial Markets, WP 1022 (July 2015).

- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, Bank Quality, judicial efficiency and loan repayment delays in Italy, Journal of Finance, WP 1072 (July 2016).